

College of Agricultural and Life Sciences Horticultural Sciences Department

1245 Fifield Hall PO Box 110690 Gainesville, FL 32611-0690 352-273-4770 352-392-6479 Fax

July 3, 2019

Donna Pearce Assistant to the Director Southern Association of Agricultural Experiment Station Directors North Carolina State University 2501 Founders Drive, 201 Patterson Hall Box 7561 Raleigh, NC 27695-7561

Dear Ms. Pearce,

RE: Reviews of S_temp1085

Project reviews were received from Robert Myers, Janet Grabowski, Elide Valencia, and Mark Van Gessel. Technical merit scores were primarily excellent, and comments were very favorable. All four reviewers recommended approval of the project. One reviewer indicated that it would be have been helpful if we had included potential collaborators along with their expertise. The key collaborators on this project will be members of the Southern Cover Crops Council (https://southerncovercrops.org/) and the collaborator information will become available when the membership is invited to complete Appendix E. No revision of the Project Outline was requested by any of the reviewers.

Sincerely, Carle A. Chase

Carlene A. Chase, Associate Professor